Author's response to reviews

Title: Prevalence of diabetes and impaired fasting glucose among residents in Three Gorges Reservoir Region, China

Authors:

Li Qi (378695810@qq.com)
Liangui Feng (flg1962@126.com)
Xianbin Ding (xianbinding@yahoo.com.cn)
Deqiang Mao (maodq@yeah.net)
Yulin Wang (wyl@cqcdc.org)
Hongyan Xiong (hongyanxiong@126.com)

Version: 4 Date: 12 September 2014

Author's response to reviews:

Dear Rita Aguirre#

Thanks for your careful reviewing. According to the suggestion, we asked a native-English friend and found a professional language editing service to copyedit the language of our revised manuscript (marked in red).

In addition, we added each author's contribution in our manuscript (Page 16, marked in red).

We sincerely hope that our response will be met with your approval.

Thank you and best regards.

Yours sincerely,
Hongyan Xiong